\documentclass[english,11pt]{article}
\usepackage[latin1]{inputenc}
\usepackage{amsmath,amsthm,amssymb,babel}
\renewcommand{\arraystretch}{1.7}
\textwidth172mm \textheight22cm \hoffset-24mm \voffset-20mm
\newtheorem{teo}{Theorem}[section]
\newtheorem{defin}{Definition}[section]
\newtheorem{prop}{Proposition}[section]
\newtheorem{example}{Example}[section]
\newtheorem{cor}{Corollary}[section]
\newtheorem{lemma}{Lemma}[section]
\newtheorem{rem}{Remark}
\numberwithin{equation}{section}
\renewcommand{\arraystretch}{1.7}
\textwidth172mm \textheight22cm \hoffset-24mm \voffset-20mm
\font\sans=cmss10
\def\proof{\noindent{\it Proof.}\ }
\def\endproof{\hfill $\Box$\par\vskip3mm}
\def\pt#1{{\it Proof of Theorem \ref{#1}.}}
\def\eq#1{(\ref{#1})}
\def\th#1{Theorem \ref{#1}}
\def\neweq#1{\begin{equation}\label{#1}}
\def\endeq{\end{equation}}
\def\weak{\rightharpoonup}
\def\de{\partial}
\def\half{{1\over2}}
\def\RR{{\mathbb R} }
\def\NN{{\mathbb N} }
\def\ZZ{{\mathbb Z} }
\def\SS{{\mathbb S} }
\def\ep{\varepsilon}

\newcommand{\uh}{\hat{u}}
\newcommand{\ub}{\bar{u}}
\newcommand{\ubx}{\bar{u}(x)}
\newcommand{\vh}{\hat{v}}
\newcommand{\vb}{\bar{v}}
\newcommand{\vbx}{\bar{v}(x)}
\newcommand{\rangleE}{\rangle_E}

\newcommand{\beps}{\mbox{\boldmath{$\varepsilon$}}}
\newcommand{\bzero}{\mbox{\boldmath{0}}}
\newcommand{\bx}{\mbox{\boldmath{$x$}}}
\newcommand{\bD}{\mbox{\boldmath{$D$}}}
\newcommand{\bvarepsilon}{\mbox{\boldmath{$\varepsilon$}}}
\newcommand{\bu}{\mbox{\boldmath{$u$}}}
\newcommand{\bI}{\mbox{\boldmath{$I$}}}
\newcommand{\bn}{\mbox{\boldmath{$n$}}}
\newcommand{\bv}{\mbox{\boldmath{$v$}}}
\newcommand{\bsigma}{\mbox{\boldmath{$\sigma$}}}
\newcommand{\btau}{\mbox{\boldmath{$\tau$}}}
\newcommand{\bxi}{\mbox{\boldmath{$\xi$}}}
\newcommand{\bnu}{\mbox{\boldmath{$\nu$}}}
\newcommand{\bbeta}{\mbox{\boldmath{$\beta$}}}
\newcommand{\fb}{\mbox{\boldmath{$f$}}}
\newcommand{\bzeta}{\mbox{\boldmath{$\zeta$}}}


\def\supp{{\rm supp}}
\def\di{\displaystyle}
\def\ri{\rightarrow}
\def\intom{\int_\Omega}
\date{}

\title{\sc Inequality problems of  quasi-hemivariational type involving  set-valued operators and a nonlinear term }
\author{\sc Nicu\c sor Costea$\,^{a}$ and Vicen\c tiu R\u adulescu$\,^{a,b,}$\thanks{Correspondence
address: Vicen\c tiu R\u adulescu, Department of Mathematics,
University of Craiova, 200585 Craiova, Romania. E-mail: {\tt
vicentiu.radulescu@imar.ro}}\\
\small $^a\,$Institute of Mathematics ``Simion Stoilow'' of the Romanian Academy 014700 Bucharest, Romania\\
\small $^b\,$Department of Mathematics, University of Craiova, 200585 Craiova, Romania\\
\small E-mail addresses: {\tt nicusorcostea@yahoo.com} \qquad {\tt
vicentiu.radulescu@imar.ro}}


\begin{document}
\begin{thebibliography}{99}
{\footnotesize


\addtolength{\leftmargin}{0.15 in}
\setlength{\itemindent}{-0.15 in}

\bibitem{AC10} I. Andrei and N. Costea, Nonlinear hemivariational inequalities
and applications to nonsmooth Mechanics, {\it Advances in
Nonlinear Variational Inequalities} {\bf 13} (1) (2010), 1-17.

\bibitem{Ber} M. Berger, {\it Nonlinearity and Functional Analysis},
Academic Press, New York, 1977.


\bibitem{bre} H. Brezis, {\it Analyse Fonctionnelle: Th\'eorie et Applications}, Masson, Paris, 1992.


\bibitem{Browder} F.E. Browder, Nonlinear maximal monotone operators in Banach space, {\it Math. Ann.} {\bf 175} (1968), 89-113.


\bibitem{Cla} F.H. Clarke, {\it Optimization and Nonsmooth Analysis},
John Wiley \& Sons, New York, 1983.


\bibitem{CM09} N. Costea and A. Matei, Weak solutions for nonlinear antiplane
problems leading to hemivariational inequalities, {\it Nonlinear
Analysis T.M.A.} {\bf 72} (2010), 3669-3680.

\bibitem{Cos-Rad1} N. Costea and V. R\u adulescu, Existence
results for hemivariational inequalities involving relaxed
$\eta-\alpha$ monotone mappings, {\it Commun. Appl. Anal.} {\bf
13} (2009), 293-304.

\bibitem{Cos-Rad2} N. Costea and V. R\u adulescu,
Hartman-Stampacchia results for stably pseudomonotone operators
and nonlinear hemivariational inequalities, {\it Applicable
Analysis} {\bf 89} (2) (2010), 175-188.



\bibitem{Cos10} N. Costea, Existence and uniqueness results for a
class of quasi-hemivariational inequalities, {\it J. Math. Anal.
Appl.} {\bf 373} (1) (2011), 305-311.


\bibitem{Fan} K. Fan, Some properties of convex sets related to
fixed point theorems, {\it Math. Ann.} {\bf 266} (1984), 519-537.



\bibitem{Varga1} F.  Faraci, A. Iannizzotto, P. Kup\'an, and  Cs. Varga,
Existence and multiplicity results for hemivariational inequalities with two parameters,
{\it Nonlinear Analysis T.M.A.}   {\bf 67} (9) (2007), 2654-2669.

\bibitem{Varga2} F. Faraci, A. Iannizzotto, H. Lisei, and Cs. Varga,
A multiplicity result for hemivariational inequalities, {\it J. Math. Anal. Appl.} {\bf 330} (1) (2007), 683-698.



\bibitem{Fich} G. Fichera, Problemi elettrostatici con vincoli unilaterali:
il problema di Signorini con ambigue condizioni al contorno,
 {\it Mem. Acad. Naz. Lincei} {\bf 7} (1964), 91-140.

\bibitem{gilbert} R.P. Gilbert, P.D.Panagiotopoulos, and P.M. Pardalos,
{\it From Convexity to Nonconvexity}, Nonconvex Optimiz. Appl., vol.~55, Kluwer Academic Publishers, Dordrecht, 2001.

\bibitem{G-L-T} R. Glowinski, J.L. Lions, and R. Tr\`{e}moli\'{e}res, {\it Numerical Analysis of Variational Inequalities},
North-Holland, Amsterdam, 1981.


\bibitem{G-M-D-R} D. Goeleven, D. Motreanu, Y. Dumont,  and  M. Rochdi,
{\it Variational and Hemivariational Inequalities, Theory, Methods
and Applicatications}, Volume I: Unilateral Analysis and
Unilateral Mechanics, Kluwer Academic Publishers,
Dordrecht/Boston/London, 2003.


\bibitem{G-M}  D. Goeleven and D. Motreanu,
{\it Variational and Hemivariational Inequalities, Theory, Methods
and Applicatications}, Volume II: Unilateral Problems, Kluwer
Academic Publishers, Dordrecht/Boston/London, 2003.



\bibitem{Har-Sta} G.J. Hartman and G. Stampacchia, On some
nonlinear elliptic differential equations, {\it Acta Math.} {\bf
112} (1966), 271-310.


\bibitem{Varga3} A. Krist\'{a}ly, Cs. Varga, and V. Varga,
A nonsmooth principle of symmetric criticality and variational-hemivariational inequalities,
{\it J. Math. Anal. Appl.} {\bf 325} (2007),  975-986.



\bibitem{Varga4} H. Lisei and Cs. Varga,
Some applications to variational-hemivariational inequalities of the principle
of symmetric criticality for Motreanu-Panagiotopoulos type functionals,
{\it J. Global Optim.} {\bf 36} (2006), 283-305.


\bibitem{Mig-Och-Sof} S. Mig\' orski, A. Ochal, and M. Sofonea,
Weak solvability of antiplane frictional contact problems for
elastic cylinders, {\it Nonlinear Analysis R.W.A.} {\bf 11} (2010),
172-183.

\bibitem{MOG2} S. Mig\' orski, A. Ochal, and M. Sofonea, Solvability of
dynamic antiplane frictional contact problems for viscoelastic
cylinders, {\it Nonlinear Analysis T.M.A.} {\bf 70} (2009),
3738-3748.


\bibitem{Minty} G.J. Minty, On the generalization of a direct method of the calculus of variations,
{\it Bull. Amer. Math. Soc.} {\bf 73} (1967), 315-321.

\bibitem{Mot-Pan} D. Motreanu and P.D. Panagiotopoulos, {\it Minimax
Theorems and Qualitative Properties of the Solutions of
Hemivariational Inequalities and Applications}, Kluwer Academic
Publishers, Nonconvex Optimization and its Applications, vol. 29,
Boston/Dordrecht/London, 1999.


\bibitem{Mot-Rad2} D. Motreanu and V. R\u adulescu, {\it Variational
and Non-variational Methods in Nonlinear Analysis and Boundary
Value Problems}, Kluwer Academic Publishers,
Boston/Dordrecht/London, 2003.



\bibitem{Nan-Pan} Z. Naniewicz and P.D. Panagiotopoulos, {\it
Mathematical Theory of Hemivariational Inequalities and
Applications}, Marcel Dekker, New York, 1995.

\bibitem{Pan1} P.D. Panagiotopoulos, Nonconvex energy functions.
Hemivariational inequalities and substationarity principles, {\it
Acta Mechanica} {\bf 42} (1983), 160-183.

\bibitem{Pan2} P.D. Panagiotopoulos, {\it Inequality Problems in Mechanics and Applications. Convex
and Nonconvex Energy Functions}, Birkhauser, Basel, 1985.

\bibitem{Pan3} P.D. Panagiotopoulos, {\it Hemivariational
Inequalities: Applications to Mechanics and Engineering},
Springer-Verlag, New York/Boston/Berlin, 1993.


\bibitem{Pan-Fun-Rad}  P.D. Panagiotopoulos,  M. Fundo, and V. R\u adulescu,
Existence theorems of Hartman-Stampacchia type for hemivariational
inequalities and applications, {\it J. Global Optim.} {\bf 15}
(1999), 41-54.

\bibitem{Papa} N.S. Papageorgiou and S.Th. Kyritsi-Yiallourou, {\it Handbook of Applied Analysis}, Springer, Advances in Mechanics and Mathematics, vol.~19, Dordrecht/London/New York, 2009.
    
\bibitem{pardalos} P.M. Pardalos, T.M. Rassias, and A.A. Khan, {\it
Nonlinear Analysis and Variational Problems}, Springer Optimization and Its Applications, vol.~35, Springer, (2010)

\bibitem{vrhin} V. R\u adulescu, {\it Qualitative Analysis of Nonlinear Elliptic Partial
Differential Equations}, Contemporary Mathematics and Its
Applications, vol.~6, Hindawi Publ. Corp., 2008.

}
\end{thebibliography}

\baselineskip16pt \maketitle

\noindent{\small{\bf Abstract.} The aim of this paper is to establish the existence of at least one solution for a general inequality of quasi-hemivariational type, whose solution is sought in a subset $K$ of a real Banach space $E$.
First, we prove the existence of solutions in the case of compact convex subsets and the case of bounded closed and convex subsets. Finally, the case when $K$ is the whole space is analyzed and necessary and sufficient conditions for the existence of solutions are stated.
Our proofs rely essentially on the Schauder's fixed point theorem and a version of the KKM principle due to Ky Fan \cite{Fan}.\\
\small{\bf 2010 Mathematics Subject Classification}: 47J20, 47H04, 49J53, 54C60, 47H05\\
\small{\bf Key words and phrases}:  Quasi-hemivariational inequality, Set-valued operator, Lower semicontinuous set-valued operator, Clarke's generalized gradient, Generalized monotonicity, KKM mapping.
}





\section{Introduction and preliminaries}

The study of inequality problems captured special attention in the last decades, one of the most
recent and general type of inequalities being the {\it hemivariational inequalities}.
The notion of hemivariational inequality was introduced by P.D. Panagiotopoulos at the beginning of the 1980's
(see e.g. \cite{Pan1,Pan2}) as a variational formulation for several classes of mechanical problems with nonsmooth and nonconvex energy
super-potentials. In the case of convex super-potentials, hemivariational inequalities reduce to {\it variational inequalities} which were studied
earlier by many authors (see e.g. Fichera \cite{Fich} or Hartman and Stampacchia \cite{Har-Sta}).

Having a life of almost thirty years now, the theory of hemivariational inequalities has produced an abundance of important results both in pure and applied mathematics as well as in other domains such as mechanics and engineering sciences (see e.g. the monographs \cite{gilbert, G-M-D-R, G-M, Mot-Pan, Mot-Rad2, Nan-Pan, Pan3, pardalos, vrhin}) as it allowed mathematical formulations for new classes of interesting problems (see e.g. \cite{AC10, CM09, Cos-Rad1, Cos-Rad2, Varga1, Varga2, Varga3, Varga4, Mig-Och-Sof, MOG2}).

The aim of this paper is to establish the existence of at least one solution for a general class of inequalities of quasi-hemivariational type. For the proof of the main results we shall use Schauder's fixed point theorem and a version of the well known KKM Principle due to Ky Fan \cite{Fan}.


\bigskip
For the convenience of the reader we present next some notations and preliminary
results from  functional analysis that will be used throughout the paper.
For a given Banach space $(X,\|\cdot\|_X)$ we denote by $X^\ast$
its dual space and by $\langle\cdot,\cdot\rangle_{X}$ the duality
pairing between $X^\ast$ and $X$.

We recall that a functional $\phi:X\ri \RR$ is called {\it locally
Lipschitz} if for every $u\in X$ there exists a neighborhood $U$
of $u$ and a constant $L_u>0$ such that
$$
|\phi(w)-\phi(v)|\leq L_u\|w-v\|_X,\quad \mbox{ for all }v,w\in U.
$$


\begin{defin}\label{ZZDefin1}
Let  $\phi:X\ri\RR$ be a locally Lipschitz functional. The
generalized derivative of $\phi$ at $u\in X$ in the direction
$v\in X$, denoted $\phi^0(u;v)$, is defined by
$$
\phi^0(u;v)=\limsup\limits_{\stackrel{w\ri u}{\lambda\downarrow
0}}\frac{\phi(w+\lambda v)-\phi(w)}{\lambda}.
$$
\end{defin}

\begin{lemma}\label{ZZLema1}
Let $\phi:X\ri\RR$ be locally Lipschitz of rank $L_u$ near the
point $u\in X$. Then
\begin{description}
\item (a) The function $v\mapsto \phi^0(u;v)$ is finite,
positively homogeneous, subadditive and satisfies
$$
|\phi^0(u;v)|\leq L_u\|v\|_X;
$$

\item (b) $\phi^0(u;v)$ is upper semicontinuous as a function of
$(u,v)$.
\end{description}
\end{lemma}
The proof can be found in Clarke \cite{Cla}, Proposition 2.1.1.



\begin{defin}\label{ZZDefin2}
The generalized gradient of a locally Lipschitz functional
$\phi:X\ri\RR$ at a point $u\in X$, denoted $\de\phi(u)$, is the
subset of $X^\ast$ defined by
$$ \de \phi(u)=\{\zeta\in X^\ast:\
\phi^0(u;v)\geq \langle\zeta,v\rangle_X,\ \mbox{ for all } v\in
X\}.
$$
\end{defin}

We point out the fact that for each $u\in X$ we have $\de
\phi(u)\neq \emptyset$. In order to see that it suffices to apply
the Hahn-Banach theorem (see e.g. Brezis \cite{bre}, p. 1).

The next lemma points out important properties of generalized
gradients.

\begin{lemma}\label{ZZLema2}

Let $\phi:X\ri\RR$ be locally Lipschitz of rank $L_u$ near the
point $u\in X$. Then
\begin{description}
\item (a) $\de \phi(u)$ is a convex, weak*  compact subset of
$X^\ast$ and
$$
\|\zeta\|_{X^\ast}\leq L_u, \quad\mbox{ for all
}\zeta\in\de\phi(u);
$$

\item (b) For each $v\in X$, one has
$$\phi^0(u;v)=\sup\{\langle\zeta,v \rangle_X:\
\zeta\in\de\phi(u)\}.$$
\end{description}
\end{lemma}

The proof can be found in Clarke \cite{Cla}, Proposition 2.1.2.

\begin{defin}\label{ZZdefin4}
A set-valued operator $T:X\ri 2^{Y}$ ($X$, $Y$ Hausdorff topological spaces)  is said to be lower semicontinuous
(Vietoris lower semicontinuous) at $u_0\in X$ (l.s.c. at $u_0$ for short),
if for all $V\subset Y$ open such that $T(u_0)\cap V\neq \emptyset$, we can find $U$ a
neighborhood of $u_0$ such that $T(u)\cap V\neq \emptyset$ for all $u\in U$.

If this is true at every $u_0\in X$, we say that $T$ is
lower semicontinuous (l.s.c. for short).
\end{defin}

It is clear from Definition \ref{ZZdefin4}, that when $A$ is single-valued,
the notion of lower semicontinuity coincides with the usual notion of
continuity of a map between two Hausdorff topological spaces.

The following proposition gives an useful characterization of lower
semicontinuity in terms of generalized sequences (see e.g. Papageorgiou and Kyritsi-Yiallourou
\cite{Papa}, p.457).

\begin{prop}\label{ZZProp2}
Given a set-valued operator $T:X\ri 2^Y$, the following statements
are equivalent:
\begin{description}

\item (a) $T$ is l.s.c.;



\item (b) If $u\in X$, $\{u_\lambda\}_{\lambda\in J}\subset X$ is a net in $X$ such that $u_\lambda\ri u$ and
$u^\ast\in T(u)$, then for each $\lambda\in J$ we can find $u_\lambda^\ast\in T(u_\lambda)$ such that $u_\lambda^\ast\ri u^\ast$ in $Y$.
\end{description}
\end{prop}



We close this section with  two theorems that will play a key role in the proof of our main results.
The first is the Schauder fixed point iheorem (for the proof see Berger \cite{Ber},  p. 90)
while the second represents a version of the KKM Principle due to Ky Fan \cite{Fan}.

\begin{teo}\label{ZZTh1} Let $X$ be a
Banach space and let $K$ be a nonempty, bounded, closed and convex
subset of $X$. Let $S : K \ri K$ be a completely continuous
operator. Then $S$ has at least one fixed point in the set $K$.
\end{teo}



\begin{teo}\label{ZZTh2}  Let
$K$ be a nonempty subset of a Hausdorff topological vector space
$X$ and let $\Theta:K\ri 2^X$ be a set-valued mapping satisfying the
following properties:
\begin{itemize}
\item  $\Theta$ is a KKM mapping;

\item  $\Theta(u)$ is closed in $X$ for every $u\in K$;

\item  there exists $u_0\in K$ such that $\Theta(u_0)$ is compact in $X$.
\end{itemize}
Then $\bigcap_{u\in K} \Theta(u)\neq \emptyset$.
\end{teo}

We recall that a set-valued mapping $\Theta:K\ri 2^X$ is said to be a
{\it KKM mapping} if for any $\{u_1,\ldots,u_n\}\subset K$, ${\rm
co}\{u_1,\ldots,u_n\}\subset \bigcup_{j=1}^n \Theta(u_j)$, where ${\rm
co}\{u_1,\ldots,u_n\}$ denotes the convex hull of
$\{u_1,\ldots,u_n\}$.


\section{Formulation of the problem}

Let $(E,\|\cdot\|_E)$ be a real Banach space which is continuously
embedded in $L^p(\Omega;\RR^n)$, for some $1 < p < +\infty$ and
$n\geq 1$, where $\Omega$ is a bounded domain in $\RR^m$, $m\geq
1$. Let $i$ be the canonical injection of $E$ into
$L^p(\Omega;\RR^n)$ and denote by $i^\ast:L^q(\Omega;\RR^n)\ri
E^\ast$ the adjoint operator of $i$ ($1/p+1/q=1$).

Throughout this paper $A:E\ri 2^{E^\ast}$ is a nonlinear set-valued
mapping, $F:E\ri E^\ast$ is a nonlinear operator and
$J:L^p(\Omega;\RR^n) \ri \RR$ is a locally Lipschitz functional.
We also assume that $h:E\ri \RR$ is a given nonnegative
functional.

\bigskip

The aim of this paper is to study the existence of solutions for
the following {\it quasi-hemivariational inequality}:

\noindent {\bf (P)} Find $u\in E$ and $u^*\in A(u)$ such that
$$
\langle u^*,v\rangle_E+h(u)J^0(iu;iv)\geq \langle
Fu,v\rangle_E,\quad \mbox{ for all } v\in E.
$$
The above problem is called a quasi-hemivariational inequality
because, in general, we cannot determine a function $G$ such that
$\de G(u)=h(u)\de J(u)$.


As we will see next problem {\bf (P)} can be rewritten
equivalently as an {\it inclusion} in the following way:

\noindent {($\mathcal{P}$)} Find $u\in E$ such that
$$
Fu\in A(u)+h(u)i^\ast \de J(iu).
$$
An element $u\in E$ is called a solution of ($\mathcal{P}$) if
there exist $u^\ast\in A(u)$ and $\zeta\in \de J(iu)$ such that
\begin{equation}\label{ZZeqn1}
\langle u^\ast,v\rangle_E +h(u)\langle
i^\ast\zeta,v\rangle_E=\langle Fu,v\rangle_E,\quad \mbox{ for all }
v\in E.
\end{equation}


\begin{prop}\label{ZZProp1}
An element $u\in E$ is a solution of problem ($\mathcal{P}$) if
and only if it solves problem {\bf (P)}.
\end{prop}
\begin{proof}
\begin{description}


\item {\bf (P)}$\Rightarrow$($\mathcal{P}$). Let $u\in E$ be a
solution of {\bf (P)}. Lemma \ref{ZZLema2} implies that there
exists $\zeta_u\in \de J(iu)$ such that for all $w\in L^p(\Omega;\RR^n)$ we have
$$ J^0(iu;w)=\langle
\zeta_u,w\rangle_{L^q\times L^p}=\sup\left\{\langle
\zeta,w\rangle_{L^q\times L^p}:\ \zeta\in \de J(iu)\right\}.
$$
Taking $w=iv$ and using the fact that $u$ is a solution of {\bf (P)} we obtain that
$$
\langle u^*,v\rangle_E+h(u)\langle i^\ast\zeta_u,v \rangle_E\geq
\langle Fu,v\rangle_E,\quad \mbox{ for all } v\in E.
$$
Taking $-v$ instead of $v$ in the above relation we deduce that
(\ref{ZZeqn1}) holds therefore $u$ is a solution of problem
($\mathcal{P}$).


\item ($\mathcal{P}$)$\Rightarrow${\bf (P)}. Let $u\in E$ be a
solution of $\mathcal{P}$. Then, there exist $u^\ast\in A(u)$ and
$\zeta\in \de J(iu)$ such that (\ref{ZZeqn1}) takes place. As
$\zeta \in\de J(iu)$ we obtain that
$$
\langle \zeta,w\rangle_{L^q\times L^p}\leq J^0(iu;w),\quad \mbox{
for all } w\in L^p(\Omega;\RR^n).
$$
For a fixed $v\in E$ we define $w=iv$ and taking into account that
$h$ is nonnegative we get
\begin{equation}\label{ZZeqn2}
h(u)\langle i^\ast\zeta,v\rangle_E=h(u)\langle
\zeta,iv\rangle_{L^q\times L^p}\leq h(u)J^0(iu;iv)
\end{equation}
Combining (\ref{ZZeqn1}) and (\ref{ZZeqn2}) we obtain that $u$
solves inequality problem {\bf (P)}.

\end{description}
\end{proof}


Sometimes, due to some technical reasons, it is useful to study
hemivariational inequalities of the type {\bf (P)} whose solution
is sought in a nonempty, closed and convex subset $K$ of $E$. This leads us
to the study of the following inequality problem:

\smallskip

\noindent ${\bf (P_K)}$ Find $u\in K$ and $u^\ast\in A(u)$ such
that
$$
\langle u^\ast,v-u\rangle_E+h(u)J^0(iu;iv-iu)\geq \langle
Fu,v-u\rangle_E,\quad\mbox{ for all } v\in K.
$$
We point out the fact that, unlike problem {\bf (P)}, the above
problem cannot be rewritten as an inclusion and this is one of the
reasons for which we prefer the hemivariational approach. However,
the formulation in terms of hemivariational inequalities has a
great advantage: that the hemivariational inequalities express a
physical principle, the principle of {\it virtual work or power}.

\medskip

\noindent We shall study three cases regarding the set $K$:
\begin{enumerate}

\item $K$ is a nonempty, compact and convex subset of the space $E$;

\item $K$ is a nonempty, bounded, closed and convex subset of the space $E$;

\item $K$ is an unbounded, closed and convex subset of $E$ (for simplicity we
shall consider that $K$ is the the whole space $E$; in this case problems
${\bf (P_K)}$ and {\bf (P)} are one and the same).
\end{enumerate}

\medskip

\noindent The novelty of our inequality problem consists in the following things:
\begin{itemize}

\item the operator $A$ is multi-valued;

\item in order to prove the existence of at least one solution in the case of bounded,
closed and convex sets we ask $A$ not to be monotone (as in most papers dealing with
hemivariational inequalities), but to be relaxed $\alpha$ monotone which is rather a
weak condition compared to monotonicity;

\item the presence of the nonlinear term in the right-hand side of the inequality which
depends on the unknown variable $u$;

\item it is a general inequality since it contains several particular cases which
lead to various known inequalities arising in many fields such as mechanics,
engineering sciences, numerical analysis.
\end{itemize}

In order to highlight the generality of our inequality problem we present below several particular
cases.
\begin{description}

\item {\it Case 1.} {\bf The operator $A$ is multi-valued.}
\begin{description}
\item {1.a)} $F\equiv 0$ and $h\equiv 0$. In this case problem ${\bf (P_K)}$ becomes

    \noindent Find $u\in K$ and $u^\ast\in A(u)$ such that
    $$
    \langle u^\ast,v-u\rangle_E\geq 0,\quad\mbox{ for all } v\in K,
    $$
    which is called the {\it generalized variational inequality } (see e.g. Minty \cite{Minty}  or Browder \cite{Browder});

\item 1.b) $Au=\de_C \phi(u)$ and $h\equiv 0$, where $\phi:E\ri (-\infty,+\infty]$ is a proper, convex functional and $\de_C:E\ri 2^{E^\ast}$  is the convex subdifferential of $\phi$, i.e.
    $$
    \de_C \phi(u)=\{\eta\in E^\ast:\ \phi(v)-\phi(u)\geq \langle\eta,v-u\rangle_E,\mbox{ for all }v\in E\}.
    $$
    In this case problem ${ \bf (P_K)}$ becomes

    \noindent Find $u\in K$ such that
    $$
    \langle-Fu,v-u\rangle_E+\phi(v)-\phi(u)\geq 0,\quad\mbox{ for all }v\in K,
    $$
    which is called the {\it mixed variational inequality } (see e.g. Glowinski, Lions and Tr\`{e}moli\'{e}res \cite{G-L-T});

\item 1.c) $Au=\de W(u)+\de_C\phi(u)$,  where $W:E\ri \RR$ is a locally Lipschitz functional and $\phi:E\ri (-\infty,+\infty]$ is a proper, convex functional. In this case problem ${\bf (P_K)}$ becomes
    $$
    W^0(u;v-u)+\phi(v)-\phi(u)+h(u)J^0(iu;iv-iu)\geq \langle Fu,v-u\rangle_E,\quad\mbox{ for all }v\in K,
    $$
    which is called {\it general quasi-hemivariational inequality} (see Costea \cite{Cos10});
\end{description}

\item {\it Case 2.} {\bf The operator $A$ is single-valued.}
\begin{description}

\item 2.a) $h\equiv 0$ and $F\equiv 0$. In this case problem ${\bf (P_K)}$ becomes

\noindent Find $u\in K$ such that
    $$
    \langle Au,v-u\rangle_E\geq 0,\quad\mbox{ for all } v\in K,
    $$
    which is called the {\it standard variational inequality } (see e.g. Hartman and Stampacchia \cite{Har-Sta});

\item 2.b) $h\equiv 1$ and $F\equiv 0$. In this case problem ${\bf (P_K)}$ becomes

\noindent Find $u\in K$  such that
    $$
    \langle Au,v-u\rangle_E+J^0(iu;iv-iu)\geq 0,\quad\mbox{ for all } v\in K,
    $$
    which is called the {\it Hartman-Stampacchia hemivariational inequality } (see Panagiotopoulos, Fundo and R\u{a}dulescu \cite{Pan-Fun-Rad});

\item 2.c) $F\equiv 0$. In this case problem ${\bf (P_K)}$ becomes

\noindent Find $u\in K$ such that
    $$
    \langle Au,v-u\rangle_E+h(u)J^0(iu;iv-iu)\geq 0,\quad\mbox{ for all } v\in K,
    $$
    which is called the {\it standard quasi-hemivariational inequality } (see e.g. Naniewicz and Panagiotopoulos \cite{Nan-Pan});
\end{description}
\end{description}


In conclusion, we do not deal with a classical hemivariational inequality and consequently several difficulties occur
in determining the existence of solutions since the classical methods fail to be applied directly.


\section{Main results}

The first main result of this paper is given by the following theorem.

\begin{teo}\label{ZZTh3}
Let $K$ be a nonempty compact convex subset of the real Banach
space $E$. Assume that:
\begin{itemize}
\item  $A:E\ri 2^{E^\ast}$ is l.s.c. with respect to the weak* topology of $E^\ast$;

\item $h:E\ri \RR$ is a continuous nonnegative functional;

\item $F:E\ri E^\ast$ is an operator such that $\limsup\limits_{n\ri\infty}\langle Fu_n,v-u_n\rangle_E\geq \langle Fu,v-u\rangle_E$, whenever $u_n\ri u$.
\end{itemize}
Then the inequality problem ${\bf (P_K)}$ has at least one
solution.
\end{teo}
\begin{proof}
Arguing by contradiction, let us assume that problem ${\bf (P_K)}$
has no solution.  Then, for each $u\in K$, there exists $v\in K$
such that
\begin{equation}\label{ZZeqn3}
\sup\limits_{u^\ast\in A(u)} \langle
u^\ast,v-u\rangle_E+h(u)J^0(iu;iv-iu)< \langle Fu,v-u\rangle_E.
\end{equation}
We introduce the set-valued mapping $\Lambda:K\ri 2^K$ defined by
$$
\Lambda(v)=\left\{ u\in K:\ \inf\limits_{u^\ast\in A(u)} \langle
u^\ast,v-u\rangle_E+h(u)J^0(iu;iv-iu)\geq \langle
Fu,v-u\rangle_E\right\}.
$$

\begin{description}
\item {\sc Claim 1.} The set $\Lambda(v)$ is nonempty and closed for each $v\in
K$.


The fact that $\Lambda(v)$ is nonempty is obvious as $v\in\Lambda(v) $ for each $v\in K$.


In order to prove the above claim let us fix $v\in K$ and consider
a sequence $\{u_n\}_{n\geq 1}\subset \Lambda(v)$ which converges
to some $u\in K$. We shall prove that $u\in \Lambda(v)$. As
$u_n\in \Lambda(v)$, for each $n\geq 1$ we get that
\begin{equation}\label{ZZeqn4}
\langle u_n^\ast,v-u_n\rangle_E+h(u_n)J^0(iu_n;iv-iu_n)\geq \langle
Fu_n,v-u_n\rangle_E,\quad \mbox{ for all } u_n^\ast\in A(u_n).
\end{equation}
Let $u^\ast\in A(u)$ be fixed and let $\bar{u}_n^\ast\in A(u_n)$
such that $\bar{u}_n^\ast \rightharpoonup u^\ast$ in $E^\ast$
(the existence of such a sequence is ensured by Proposition \ref{ZZProp2}
and the fact that  $A$ is l.s.c. with respect to the
weak* topology of $E^\ast$). On the
other hand, using the continuous embedding of $E$ into
$L^p(\Omega;\RR^n)$ we obtain that $iu_n\ri iu$ in
$L^p(\Omega;\RR^n)$. Passing to $\limsup$ as $n\ri \infty$ in
(\ref{ZZeqn4}) we obtain the following estimates:
\begin{eqnarray*}
\langle Fu,v-u\rangle_E & \leq & \limsup\limits_{n\ri\infty}\langle
Fu_n,v-u_n\rangle_E\\
 & \leq & \limsup\limits_{n\ri\infty} \left[\langle
 \bar{u}_n^\ast,v-u_n\rangle_E+h(u_n)J^0(iu_n;iv-iu_n)\right]\\
 & \leq & \limsup\limits_{n\ri\infty} \langle
 \bar{u}_n^\ast,v-u_n\rangle_E+\limsup\limits_{n\ri\infty}
 \left[h(u_n)-h(u)+h(u)\right]J^0(iu_n;iv-iu_n)\\
 & \leq & \langle u^\ast,v-u\rangle_E+\limsup\limits_{n\ri\infty}
 \left[h(u_n)-h(u)\right]J^0(iu_n;iv-iu_n)\\
 &  &+\limsup\limits_{n\ri\infty}h(u)J^0(iu_n;iv-iu_n)\\
 & \leq & \langle u^\ast,v-u\rangle_E+h(u)J^0(iu;iv-iu).
\end{eqnarray*}
This shows that $u\in\Lambda(v)$ hence $\Lambda(v)$ is a closed
set and the proof of the claim is now complete.
\end{description}



According to (\ref{ZZeqn3}) for each $u\in K$ there exists $v\in
K$ such that $u\in [\Lambda(v)]^c=E-\Lambda(v)$. This means that
the family $\{[\Lambda(v)]^c\}_{v\in K}$ is an open covering of
the compact set $K$. Therefore there exists a finite subset
$\{v_1,\ldots,v_N\}$ of  $K$ such that
$\{[\Lambda(v_j)]^c\}_{1\leq j\leq N}$ is a finite subcover of $K$. For
each $j\in\{1,\ldots,N\}$ let $\delta_j(u)$ be the distance between $u$
and the set $\Lambda(v_j)$ and define $\beta_j:K\ri\RR$ as follows:
$$
\beta_j(u)=\frac{\delta_j(u)}{\sum\limits_{k=1}^N\delta_k(u)}.
$$
Clearly, for each $j\in\{1,\ldots,N\}$, $\beta_j$ is a Lipschitz continuous function
that vanishes on $\Lambda(v_j)$ and
$0\leq \beta_j(u)\leq 1$, for all $u\in K$. Moreover, $\sum_{j=1}^N \beta_j(u)=1$.
Let us consider next the operator $S:K\ri K$ defined by
$$
S(u)=\sum_{j=1}^N \beta_j(u)v_j.
$$
We shall prove that $S$ is a completely continuous operator. We have
\begin{eqnarray*}
\|Su_1-Su_2\|_E & = & \left\| \sum\limits_{j=1}^N  (\beta(u_1)-\beta(u_2)) v_j \right\|_E\\
                & \leq & \sum\limits_{j=1}^N \|v_j\|_E \;\left\| \beta(u_1)-\beta(u_2) \right\|_E\\
                & \leq & \sum\limits_{j=1}^N \|v_j\|_E\; L_j\; \|u_1-u_2\|_E\\
                &  \leq & L\; \|u_1-u_2\|_E,
\end{eqnarray*}
which shows that $S$ is Lipschitz continuous hence continuous.

Let $M$ be a bounded subset of $K$. As $\overline{S(M)}$ is a closed subset of the compact set
$K$ we conclude that $S(M)$ is relatively compact, hence $S$ maps
bounded sets into relatively compact sets which shows that $S$ is
a compact map. Thus, by Schauder's fixed point theorem, there
exists $u_0\in K$ such that $S(u_0)=u_0$.

Let us define next the functional $g:K\ri \RR$
$$
g(u)=\inf\limits_{u^\ast\in A(u)}\langle
u^\ast,S(u)-u\rangle_E+h(u)J^0(iu,iS(u)-iu)-\langle
Fu,S(u)-u\rangle_E.
$$
Taking into account Lemma \ref{ZZLema1} and the way the operator
$S$ was constructed, for each $u\in K$, we have:
\begin{eqnarray*}
g(u) & = & \inf\limits_{u^\ast\in A(u)}\left\langle
u^\ast,\sum_{j=1}^N
\beta_j(u)(v_j-u)\right\rangle_E+h(u)J^0\left(iu,\sum_{j=1}^N\beta_j(u)(iv_j-iu)\right)\\
    &  &-\left\langle Fu,\sum_{j=1}^N\beta_j(u)(v_j-u)\right\rangle_E\\
    & \leq & \sum_{j=1}^N\beta_j(u)\left[\inf\limits_{u^\ast\in A(u)}\langle u^\ast,v_j-u\rangle_E+h(u)J^0(iu,iv_j-iu)-\langle
Fu,v_j-u\rangle_E\right].
\end{eqnarray*}
Let $u\in K$ be arbitrary fixed. For each index $j\in\{1,\ldots,N\}$ we distinguish the following possibilities:
\begin{description}

\item $\bullet$ $u\in\left[\Lambda(v_j)\right]^c$. In this case we have
$$
\beta_j(u)>0
$$
and
$$
\inf\limits_{u^\ast\in A(u)}\langle u^\ast,v_j-u\rangle_E+h(u)J^0(iu,iv_j-iu)-\langle
Fu,v_j-u\rangle_E<0.
$$

\item $\bullet$ $u\in \Lambda(v_j)$. In this case we have
$$
\beta_j(u)=0
$$
and
$$
\inf\limits_{u^\ast\in A(u)}\langle u^\ast,v_j-u\rangle_E+h(u)J^0(iu,iv_j-iu)-\langle
Fu,v_j-u\rangle_E\geq 0.
$$

\end{description}

Taking into account that $K\subseteq \cup_{j=1}^N \left[\Lambda(v_j)\right]^c$ we deduce that there exists at least one index $j_0\in\{1,\ldots,N\}$ such that $u\in \left[\Lambda(v_{j_0})\right]^c$. This shows that $g(u)<0$ for all $u\in K$.

On the other hand, $g(u_0)=0$ and thus we have obtained a
contradiction that completes the proof.

\end{proof}


We point out the fact that in the above case when $K$ is a compact convex subset of $E$ we do not impose any monotonicity conditions on $A$, nor we
assume $E$ to be a reflexive space. However, in applications, most problems lead to an inequality whose solution is sought in a closed and convex subset of the space $E$. Weakening the hypotheses on $K$ by assuming that $K$ is only bounded, closed and convex, we need to impose certain monotonicity properties on $A$ and assume in addition that $E$ is reflexive. We shall use a kind of generalized monotonicity, so called {\it relaxed $\alpha$ monotonicity}.
We recall the following definition.

\begin{defin}\label{ZZdefin5}
A set-valued mapping $T:E\ri 2^{E^\ast}$ is said to be relaxed $\alpha$ monotone if there exists a functional $\alpha:E\ri \RR$ such that for all $u,v\in E$ we have
\begin{equation}\label{ZZeqnMon}
\langle v^\ast-u^\ast,v-u\rangle_E\geq \alpha(v-u),\quad \mbox{ for all } v^\ast\in T(v) \mbox{ and all } u^\ast\in T(u).
\end{equation}
\end{defin}


\begin{description}
\item { \sc Special cases.}

\item \quad $\bullet$ If $\alpha(u)=m\|u\|_E^2$, with $m>0$ constant, then \eq{ZZeqnMon} becomes
$$
\langle v^\ast-u^\ast,v-u\rangle_E\geq m\|v-u\|^2_E,\quad \mbox{ for all } v^\ast\in T(v) \mbox{ and all } u^\ast\in T(u),
$$
and $T$ is said to be {\it strongly monotone; }

\item \quad $\bullet$ If $\alpha(u)\equiv m$, with $m>0$ constant, then \eq{ZZeqnMon} becomes
$$
\langle v^\ast-u^\ast,v-u\rangle_E\geq m>0,\quad \mbox{ for all $u\not=v$},\  v^\ast\in T(v),\  u^\ast\in T(u),
$$
and $T$ is said to be {\it strictly monotone;}

\item \quad $\bullet$ If $\alpha(u)\equiv 0$, then \eq{ZZeqnMon} becomes
$$
\langle v^\ast-u^\ast,v-u\rangle_E\geq 0,\quad \mbox{ for all } v^\ast\in T(v) \mbox{ and all } u^\ast\in T(u),
$$
and $T$ is said to be {\it monotone; }

\item \quad $\bullet$ If $\alpha(u)=-m\|u\|_E^2$, with $m>0$ constant, then \eq{ZZeqnMon} becomes
$$
\langle v^\ast-u^\ast,v-u\rangle_E\geq -m\|v-u\|^2_E,\quad \mbox{ for all } v^\ast\in T(v) \mbox{ and all } u^\ast\in T(u),
$$
and $T$ is said to be {\it relaxed monotone.}
\end{description}
From the above definitions, we have the following implications (and the inverse of every
implication is not true):
\begin{center}
\begin{tabular}{|c|c|c|c|c|c|c|c|c|}
  \hline
   {\footnotesize strongly monotone }& $\Rightarrow$ & {\footnotesize strictly monotone} & $\Rightarrow$ & {\footnotesize monotone} & $\Rightarrow$ & {\footnotesize relaxed monotone} & $\Rightarrow$ & {\footnotesize relaxed $\alpha$ monotone}\\
  \hline
\end{tabular}
\end{center}

We are now able to formulate another main result concerning the existence of solutions on bounded, closed and convex subsets.

\begin{teo}\label{ZZTh4}
Let $K$ be a nonempty, bounded, closed and convex subset of the
real reflexive Banach space $E$ which is compactly embedded in $L^p(\Omega;\RR^n)$. Assume that:
\begin{itemize}

\item $A:E\ri 2^{E^\ast}$ is l.s.c. with respect to the weak topology of $E^\ast$ and
relaxed $\alpha$ monotone;

\item $\alpha:E\ri \RR$ is a functional such that $\limsup\limits_{n\ri\infty}\alpha(u_n)\geq \alpha(u)$
whenever $u_n\rightharpoonup u$ and  $\lim\limits_{t\downarrow 0} \frac{\alpha(tu)}{t}=0$;

\item $h:E\ri \RR$ is a nonnegative sequentially weakly continuous functional;

\item $F:E\ri E^\ast$ is an operator such that the application
$u\mapsto \langle Fu,v-u\rangle_E$ is weakly lower semicontinuous.
\end{itemize}
Then the inequality problem ${\bf (P_K)}$ has at least one solution
in $K$.
\end{teo}

\begin{proof}
Let us define the set-valued mapping $\Theta:K\ri 2^K$
$$
\Theta(v)=\left\{u\in K:\ \inf\limits_{v^\ast\in A(v)} \langle
v^\ast,v-u\rangle_E+h(u)J^0(iu;iv-iu)- \langle
Fu,v-u\rangle_E\geq\alpha(v-u) \right\}.
$$

\begin{description}
\item  {\sc Claim 2.} The set $\Theta(v)$ is weakly closed for
each $v\in K$.

In order to prove the above claim let us fix $v\in K$ and consider
a sequence $\{u_n\}_{n\geq 1}\subset \Theta(v)$ such that
$u_n\rightharpoonup u$ in $E$. We must prove that
$u\in\Theta(v)$. First we observe that the compactness of the embedding
operator $i$ implies that the sequence $\{iu_n\}_{n\geq 1}$ converges
strongly to $iu$ in $L^p(\Omega,\RR^n)$.

For each $v^\ast\in A(v)$ we have
\begin{eqnarray*}
\alpha(v-u) & \leq &  \limsup\limits_{n\ri\infty} \alpha(v-u_n) \\
    & \leq & \limsup\limits_{n\ri\infty} \left[\langle
     v^\ast,v-u_n\rangle_E+h(u_n)J^0(iu_n;iv-iu_n)-\langle Fu_n,v-u_n\rangle_E\right]\\
    & \leq & \langle v^\ast,v-u\rangle_E+h(u)J^0(iu,iv-iu)-\langle Fu,v-u\rangle_E,
\end{eqnarray*}
which shows that $u\in \Theta(v)$ and thus the proof of the claim
is complete.
\end{description}

\begin{description}
\item {\sc Claim 3.} $\Theta$ is a KKM mapping.


Arguing by contradiction let us assume that $\Theta$ is not a KKM mapping.
According to the definition of a KKM mapping there exists a finite
subset $\{v_1,\ldots,v_N\}\subset K$ and $u_0=\sum_{j=1}^N
\lambda_j v_j$, with $\lambda_j\in [0,1]$ and
$\sum_{j=1}^N\lambda_j=1$ such that $u_0\not\in \bigcup_{j=1}^N
\Theta(v_j)$. This is equivalent to
\begin{equation}\label{ZZeqn5}
\inf\limits_{v_j^\ast\in A(v_j)}\langle v_j^\ast,
v_j-u_0\rangle_E+h(u_0)J^0(iu_0;iv_j-iu_0)- \langle
Fu_0,v_j-u_0\rangle_E<\alpha(v_j-u_0),
\end{equation}
for all $j\in\{1,\ldots,N\}$.

On the other hand, $A$ is a relaxed $\alpha$ monotone operator and thus, for each $j\in\{1,\ldots,N\}$ we have
\begin{equation}\label{ZZeqn6}
\langle u_0^\ast-v_j^\ast,v_j-u_0\rangle_E \leq -\alpha(v_j-u_0),\quad \mbox{ for all } u_0^\ast\in A(u_0) \mbox{ and all } v_j^\ast\in A(v_j).
\end{equation}

Combining \eq{ZZeqn5} and \eq{ZZeqn6} we are led to
\begin{equation}\label{ZZeqn11}
 \langle u_0^\ast, v_j-u_0 \rangle_E+h(u_0)J^0(iu_0;iv_j-iu_0)- \langle
Fu_0,v_j-u_0\rangle_E<0,\quad \mbox{ for all } u_0^\ast\in A(u_0).
\end{equation}

Using \eq{ZZeqn11} and the fact that $J^0(iu_0;\cdot)$ is subadditive (see Lemma \ref{ZZLema1}), for a fixed $u_0^\ast\in A(u_0)$ we have
\begin{eqnarray*}
0   & = & \langle u_0^\ast,u_0-u_0\rangle_E+h(u_0)J^0(iu_0;iu_0-iu_0) - \langle Fu_0,u_0-u_0\rangle_E\\
    & = & \left\langle u_0^\ast,\sum_{j=1}^N\lambda_j(v_j-u_0)\right\rangle_E+h(u_0)J^0\left(iu_0;\sum_{j=1}^N\lambda_j(iv_j-iu_0)\right) - \left\langle Fu_0,\sum_{j=1}^N\lambda_j(v_j-u_0)\right\rangle_E\\
    & \leq & \sum_{j=1}^N \lambda_j\left[\langle u_0^\ast,v_j-u_0\rangle_E+h(u_0)J^0(iu_0;iv_j-iu_0) - \langle Fu_0,v_j-u_0\rangle_E\right]\\
    & < & 0,
\end{eqnarray*}
which obviously is a contradiction and thus the proof of the claim is complete.

\end{description}



We already know from Claim 2 that $\Theta(v)$ is a weakly closed subset of
$K$, for each $v\in K$. On the other hand, $K$ is a weakly compact set as it is a
bounded, closed and convex subset of the real reflexive Banach
space $E$.  Therefore $\Theta(v)$ it is weakly compact for each
$v\in K$. Thus we can apply the KKM Principle to conclude that
$\bigcap_{v\in K} \Theta(v)\neq \emptyset$.

Let $u_0\in\bigcap_{v\in K} \Theta(v)$. This implies that for
each $w\in K$ we have
$$
\inf\limits_{w^\ast\in A(w)}\langle w^\ast, w-u_0
\rangle_E+h(u_0)J^0(iu_0;iw-iu_0)-\langle
Fu_0,w-u_0\rangle_E\geq\alpha(w-u_0).
$$

Let $v\in K$ be fixed and define $w_\lambda=u_0+\lambda(v-u_0)$,
$\lambda\in (0,1)$.  Using the fact that $w_\lambda\in K$ and
taking into account the above relation and Lemma \ref{ZZLema1} we
deduce that
$$
\langle w_\lambda^\ast,v-u_0\rangle_E+ h(u_0)J^0(iu_0,iv-iu_0)-\langle
Fu_0,v-u_0\rangle_E\geq\frac{ \alpha(\lambda(v-u_0))}{\lambda},\mbox{ for all
} w_\lambda^\ast\in A(w_\lambda).
$$
Letting $\lambda\ri 0$ and using the l.s.c. of $A$
we obtain that $u_0$ solves problem ${\bf (P_K)}$.



\end{proof}



As we have seen above the boundedness of the set $K$ played a key role in proving that problem ${\bf (P_K)}$
admits at least one solution. In the case when $K$ is the whole space $E$, assuming that the same hypotheses
as in Theorem \ref{ZZTh4} hold, we shall need an extra condition to overcome the lack of boundedness. For each real
number $R>0$ taking $K=\bar{B}(0;R)=\{u\in E:\ \|u\|_E\leq R\}$ we know from  Theorem \ref{ZZTh4} that problem


\bigskip
\noindent ${\bf (P_R)}$ Find $u_R\in \bar{B}(0;R)$ and $u_R^\ast\in A(u_R)$ such that
$$
\langle u_R^\ast,v-u_R\rangle_E+h(u_R)J^0(iu_R;iv-iu_R)\geq\langle Fu_R,v-u_R\rangle_E,\quad \mbox{ for all } v\in \bar{B}(0;R),
$$
admits at least one solution.


\begin{teo}\label{ZZTh5}
Assume that the same hypotheses as in Theorem \ref{ZZTh4}
hold in the case $K=E$. Then problem ${\bf (P)}$ admits at least one solution if and only if the following condition holds true:
\begin{itemize}
\item There exists $R>0$ such that at least one solution $u_R$ of problem ${\bf (P_R)}$ satisfies $u_R\in {\rm int\;} \bar{B}(0;R)$.
\end{itemize}
\end{teo}

\begin{proof}
The necessity is obvious.

In order to prove the sufficiency let us fix $v\in E$. We shall prove that $u_R$ is a solution of ${\bf (P)}$. First we define
$$
\lambda=\left\{
\begin{array}{ll}
1 & \mbox{ if } u_R=v\\
\frac{R-\|u_R\|_E}{\|v-u_R\|_E} & \mbox{ otherwise }.
\end{array}
\right.
$$

Since $u_R\in {\rm int\;}B(0;R)$ we conclude that $\lambda>0$ and that $w_\lambda= u_R+\lambda (v-u_R)\in \bar{B}(0;R)$. Using that $u_R$ solves
problem ${\bf (P_R)}$ we find
\begin{eqnarray*}
\langle Fu_R,\lambda(v-u_R)\rangle_E & = &\langle Fu_R,w_\lambda-u_R\rangle_E \\
    & \leq &\langle u_R^\ast,w_\lambda-u_R)\rangle_E+h(u_R)J^0(iu_R;iw_\lambda-iu_R)\\
    & = & \langle u_R^\ast,\lambda(v-u_R)\rangle_E+h(u_R)J^0(iu_R;\lambda(iv-iu_R))\\
    & = &\lambda\left[\langle u_R^\ast,v-u_R\rangle_E+h(u_R)J^0(iu_R;iv-iu_R)\right].
\end{eqnarray*}
Dividing by $\lambda>0$ we conclude that $u_R$ solves problem ${\bf (P)}$.
\end{proof}

\begin{cor}\label{ZZCorol1}
Let us assume that the same hypotheses as in Theorem \ref{ZZTh4}
hold in the case $K=E$. Then a sufficient condition for problem {\bf (P)} to admit a solution is:
\begin{itemize}
\item There exists $R_0>0$ such that for each $u\in E\setminus \bar{B}(0;R_0)$ there exists $v\in {\rm int\;} \bar{B}(0;R_0)$ with the property that
$$
\sup\limits_{u^\ast\in A(u)}\langle u^\ast,v-u\rangle_E+h(u)J^0(iu;iv-iu)<\langle Fu,v-u\rangle_E.
$$
\end{itemize}
\end{cor}

\begin{proof}
Let us fix $R>R_0$. According to Theorem \ref{ZZTh4} there exists $u_R\in \bar{B}(0,R)$ and $\bar{u}_R^\ast\in A(u_R)$ such that
\begin{equation}\label{ZZeqn9}
\langle \bar{u}_R^\ast,v-u_R\rangle_E+h(u_R)J^0(iu_R;iv-iu_R)\geq \langle Fu_R,v-u_R\rangle_E,\quad \mbox{ for all } v\in\bar{B}(0;R) .
\end{equation}
\begin{description}
\item {\it Case 1.} $u_R\in {\rm int \;} \bar{B}(0;R)$.

In this case we have nothing to prove, Theorem \ref{ZZTh5} showing that $u_R$ is a solution of problem ${\bf (P)}$.

\item {\it Case 2.} $u_R\in \partial \bar{B}(0;R)$.

In this case $\|u_R\|_E=R>R_0$ and thus $u_R\in E\setminus\bar{B}(0;R_0)$. According to our hypothesis there exists $\bar{v}\in{\rm int\;}\bar{B}(0;R_0)$ such that
\begin{equation}\label{ZZeqn110}
\sup\limits_{u_R^\ast\in A(u_R)}\langle u_R^\ast,\bar{v}-u_R\rangle_E+h(u_R)J^0(iu_R;i\bar{v}-iu_R)<\langle Fu_R,\bar{v}-u_R\rangle_E.
\end{equation}

Let us fix $v\in E$. Defining
$$
\lambda=\left\{
\begin{array}{ll}
1 & if v=\bar{v}\\
\frac{R-R_0}{\|v-\bar{v}\|_E} & otherwise,
\end{array}
\right.
$$
we observe that $w_\lambda = \bar{v}+\lambda(v-\bar{v})\in\bar{B}(0;R)$.
On the other hand we observe that
\begin{eqnarray*}
w_\lambda-u_R & = & \bar{v}-u_R+\lambda(v-\bar{v})+\lambda u_R -\lambda u_R\\
    & = & \lambda (v-u_R)+(1-\lambda)(\bar{v}-u_R).
\end{eqnarray*}
Taking $w_\lambda$ instead of $v$ in \eq{ZZeqn9} and using \eq{ZZeqn110} we are led to the following estimates
\begin{eqnarray*}
\left\langle Fu_R,\lambda (v-u_R)+(1-\lambda)(\bar{v}-u_R)\right\rangle_E  & = & \langle Fu_R,w_\lambda-u_R\rangle_E  \\
    & \leq & \langle \bar{u}_R^\ast,w_\lambda-u_R\rangle_E +h(u_R)J^0(iu_R;iw_\lambda-iu_R)\\
    & \leq & \lambda \left[\langle \bar{u}_R^\ast,v-u_R\rangle_E +h(u_R)J^0(iu_R;iv-iu_R)\right]\\
    &  & +(1-\lambda)\left[\langle \bar{u}_R^\ast,\bar{v}-u_R\rangle_E +h(u_R)J^0(iu_R;i\bar{v}-iu_R)\right]\\
    & \leq & \lambda \left[\langle \bar{u}_R^\ast,v-u_R\rangle_E +h(u_R)J^0(iu_R;iv-iu_R)\right]\\
    &  & +(1-\lambda) \langle Fu_R,\bar{v}-u_R\rangle_E.
\end{eqnarray*}

This shows that
$$
\langle \bar{u}_R^\ast,v-u_R\rangle_E +h(u_R)J^0(iu_R;iv-iu_R)\geq \langle Fu_R;v-u_R\rangle_E,\quad \mbox{ for all }v\in E,
$$
which means that $u_R$ solves problem {\bf (P)} and thus the proof is complete.

\end{description}

\end{proof}

\begin{cor}\label{ZZCorol2}
Let us assume that the same hypotheses as in Theorem \ref{ZZTh4}
hold in the case $K=E$. Assume in addition that:
\begin{itemize}

\item $A$ is coercive, i.e. there exists a function $c:\RR_+\ri\RR_+$ with the property that
$\lim\limits_{r\ri \infty}c(r)=+\infty$ such that
$$
\inf\limits_{u^\ast\in A(u)}\langle u^\ast,u\rangle_E\geq c(\|u\|_E)\|u\|_E;
$$

\item there exists a constant $k>0$ such that $h(v)J^0(iv;-iv)\leq k\|v\|_E$ for all $v\in E$;

\item there exists a constant $m>0$ such that $\|Fu\|_{E^\ast}\leq m$ for all $u\in E$;
\end{itemize}
Then the inequality problem {\bf (P)} has at least one solution.
\end{cor}



\begin{proof}
For each $R>0$ Theorem \ref{ZZTh4} guarantees that there exist $u_R\in E$ and $u_R^\ast\in A(u_R)$ such that
\begin{equation}\label{ZZeqn10}
\langle u_R^\ast,v-u_R\rangle_E+h(u_R)J^0(iu;iv-iu_R)\geq \langle Fu_R,v-u_R\rangle_E, \quad\mbox{ for all } v\in \bar{B}(0;R).
\end{equation}

We shall prove that there exists $R_0>0$ such that $u_{R_0}\in {\rm int\;} \bar{B}(0;R_0)$.  According to Theorem \ref{ZZTh5}, this is equivalent to the fact that $u_R$ is a solution of problem {\bf (P)}.

Arguing by contradiction let us assume that $u_R\in \de\bar{B}(0;R)$ for all $R>0$. Taking $v=0$ in \eq{ZZeqn10} we have
\begin{eqnarray*}
c(R)R & = & c(\|u_R\|)\|u_R\|_E\\
      & \leq & \langle u_R^\ast, u_R\rangle_E\\
      & \leq & \langle Fu_R,u_R\rangle_E+h(u_R)J^0(iu_R;-iu_R)\\
      & \leq & \|Fu_R\|_{E^\ast}\|u_R\|_E+k\,\|u_R\|_E\\
      & = & (m+k)R.
\end{eqnarray*}
Dividing by $R>0$ we obtain that $c:\RR_+\ri\RR_+$ is bounded from above which contradicts the fact that $\lim\limits_{R\ri\infty} c(R)=+\infty$.
\end{proof}



\bigskip

\bigskip


\noindent {\bf Acknowledgements.} N.~Costea has been supported by a Bitdefender Postdoctoral Fellowship at the Institute of Mathematics ``Simion Stoilow'' of the Romanian Academy. V.~R\u adulescu acknowledges the support through  Grant CNCSIS PCCE--8/2010
{\it ``Sisteme diferen\c{t}iale \^{\i}n analiza neliniar\u{a} \c{s}i aplica\c{t}ii"}.






\end{document}
